Author’s response to reviews

Title: Clinical value of DPOC for detecting and removing residual common bile duct stones (video)

Authors:

Jun-Jie Yang (junjie1001jjy@aliyun.com)
Xiong-Chang Liu (liuxiongchang1935@163.com)
Xiao-Qin Chen (chenxiaoqin03cxq@sohu.com)
Qi-Yong Zhang (z297qiyong@sohu.com)
Tian-Rang Liu (tianrang197liu@sohu.com)

Version: 3  Date: 09 Jul 2019

Author’s response to reviews:

Comments 1 In Funding, please state whether or not the funding body played any roles in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Reply: Thank you. We have added the sentence “The funding was used in the design of the study and collection, analysis, and interpretation of data.”.

Comments 2 Please include a statement in the Authors’ contributions section to the effect that all authors have read and approved the final manuscript, and ensure that this is the case.

Reply: Thank you. We have added the sentence “all authors have read and approved the final manuscript”.